Author's response to reviews

Title: Comparability of localization data in transnasal and transoral esophagastroduodenoscopy

Authors:

Serhat Aymaz (aymaz@kliniken-koeln.de)
Bernd Krakamp (krakampb@kliniken-koeln.de)
Oliver Kirschberg (kirschbergo@kliniken-koeln.de)
Rolf Lefering (rolf.lefering@uni-wh.de)

Version: 2 Date: 10 April 2010

Author's response to reviews:

Dear Sirs,

we would like to add the following sentence in compliance with the Helsinki Declaration:

The study was performed with the approval of the ethics committee of the University of Cologne (“Ethik Kommission der Medizinischen Fakultät der Universität zu Köln”, Ref.-no.: 05-003, 21st of February 2005).